

Reviewer 1:

Thank you for your swift review and acceptance of our manuscript. We greatly appreciate your positive feedback and endorsement of our work.

Reviewer 2:

Thank you for your insightful comments on our manuscript. We have addressed your suggestions by discussing additional factors contributing to oxalate crystal nephropathy, and expanding on future research directions. While we couldn't include polarized light microscopy examination and electron microscopy images (Due to lack of images), we emphasized the need for further comparative studies. Your feedback has strengthened our manuscript significantly.